Author's response to reviews

Title: Enhancing SHP-1 expression by 5-Azacytidine inhibits STAT3 activation and confers sensitivity in Lestaurnitinib (CEP-701) resistant FLT3-ITD positive Acute Myeloid Leukemia

Authors:

Hamid AN Al-Jamal (aljamalhamid08@gmail.com)
Siti Asmaa Mat Jusoh (asma_sof88@yahoo.com)
Rosline Hassan (roslin@usm.my)
Muhammad Farid Johan (faridjohan@usm.my)

Version: 2 Date: 4 January 2015

Author's response to reviews:

Dear Editor,

This is the revised version of the manuscript. The line and page numbers are included in the main text. No human subjects/materials/data have been carried out/reported in this research.

Thank you

MF Johan